Author's response to reviews

Title: Acoustically detected Magnetic Particle Imaging

Authors:

Bernhard Gleich (bernhard.gleich@philips.com)
Jürgen Weizenecker (juergen.weizenecker@hs-karlsruhe.de)
Jörn Borgert (joern.borgert@philips.com)

Version: 6 Date: 18 May 2011

Author's response to reviews:

Dear Ladies and Gentleman

The proposed changes to the title page have been implemented. The abstract has been rewritten to match the journal's guidelines. The whole manuscript has been transferred into the provided template to ensure matching the journal's style.

The differentiation between the "Results" and the "Discussion" section seemed a little artificial to us, so we would propose, if possible, to merge these into one section called "Results and Discussions".

Kindest regards

Jörn Borgert